Author's response to reviews

Title: The training needs of Turkish emergency department personnel regarding intimate partner violence

Authors:

H. Asli Davas Aksan (asli.davas@ege.edu.tr)
Feride Aksu (feride.aksu@ege.edu.tr)

Version: 6 Date: 13 November 2007

Author's response to reviews: see over
Dear BMC Public Health Editors:

We have done the formatting changes requested by The BioMed Central Editorial Production Team. The response is written immediately below each comment.
I look forward to hearing from you.
Yours sincerely
Aslı Davas

1-We have removed all the unnecessary capitals and the title changed as follows: “The training needs of Turkish emergency department personnel regarding intimate partner violence”

2-We super scripted the numbers linking authors affiliations.

3-We have deleted the title at the top of the abstract.

4-We have changed authors contributions part.

5-References changed. The term et al. was taken out. Reference 25 was separated.

6-We have removed the ekstra fullstop after ref.21.